Author's response to reviews

Title: Custom-designed orthopedic implants evaluated using finite element analysis of patient-specific computed tomography data: femoral-component case study

Authors:

Ola L. A. Harrysson (harrysson@ncsu.edu)
Yasser A Hosni (yhosni@mail.ucf.edu)
Jamal F Nayfeh (nayfeh@mail.ucf.edu)

Version: 6 Date: 5 September 2007

Author's response to reviews:

Dear Editor,

I have changed the legend for Figure 6 so there is only one. Please let me know if there is anything else that I need to do.

Regards,

Ola Harrysson